Author's response to reviews

Title: Early effect of oral administration of omeprazole with mosapride as compared with those of omeprazole alone on the intragastric pH

Authors:

Hiroshi Iida (hiro0905@yokohama-cu.ac.jp)
Masahiko Inamori (inamorim@med.yokohama-cu.ac.jp)
Tetsuro Fujii (tetsuro@yokohama-cu.ac.jp)
Yusuke Sekino (ysekino@yokohama-cu.ac.jp)
Hiroki Endo (endo1978@yokohama-cu.ac.jp)
Kunihiro Hosono (hiro1017@fukuhp.yokohama-cu.ac.jp)
Takashi Nonaka (nonaka_taka@yahoo.co.jp)
Tomoko Koide (tomo-k57@fukuhp.yokohama-cu.ac.jp)
Hirokazu Takahashi (hirokazu@med.yokohama-cu.ac.jp)
Masato Yoneda (yoneda@med.yokohama-cu.ac.jp)
Ayumu Goto (qq9v6m3u9@able.ocn.ne.jp)
Yasunobu Abe (a0121@yokohama-cu.ac.jp)
Noritoshi Kobayashi (norikoba@yokohama-cu.ac.jp)
Hiroyuki Kirikoshi (hkirkos@med.yokohama-cu.ac.jp)
Kensuke Kubota (kubotak@yokohama-cu.ac.jp)
Satoru Saito (ssai1423@yokohama-cu.ac.jp)
Eiji Gotoh (egotoh@xa2.so-net.ne.jp)
Shin Maeda (smaeda@med.yokohama-cu.ac.jp)
Atsushi Nakajima (nakajima-tky@umin.ac.jp)

Version: 2 Date: 19 February 2012

Author's response to reviews: see over
Dear Editors,

We enjoy reading BMC Gastroenterology for the information it contains on both clinical and research topics. We would like to re-submit the revised manuscript, entitled “Early effect of oral administration of omeprazole with mosapride as compared with those of omeprazole alone on the intragastric pH” for publication in BMC Gastroenterology. We believe that many physicians will be interested in our research. We hope that you will find our article suitable for your esteemed journal.

Yours sincerely,

Masahiko Inamori, MD., PhD.
Gastroenterology Division, Yokohama City University School of Medicine.
3-9 Fukuura, Kanazawa-ku, Yokohama, Kanagawa, 236-0004, Japan
Telephone: +81-45-787-2640   FAX: +81-45-784-3546
E-mail: inamorim@med.yokohama-cu.ac.jp